Author's response to reviews

Title: Demographic, risk behaviour and personal network variables associated with prevalent hepatitis C, hepatitis B, and HIV infection in injection drug users in Winnipeg, Canada.

Authors:

John L Wylie (JWylie@gov.mb.ca)
Lena Shah (Lena.Shah@moh.gov.on.ca)
Ann M Jolly (ann_m_jolly@phac-aspc.gc.ca)

Version: 4 Date: 8 September 2006

Author's response to reviews:

As per your instructions we have:

1. removed the full stop from the end of the title.
2. Removed the line numbers
3. Typed out each affiliation in full.
4. Used sentence case for each heading.
5. Corrected all journal abbreviations to Medline versions.
6. Used a Biomed Central journal style which I had downloaded to my Endnote program.

Thanks for accepting the article.